Author's response to reviews

Title: Use of email for patient communication in student health care: a cross-sectional study

Authors:

Johanna Castren (johanna.castren@uta.fi)
Marja Niemi (marja.niemi@stakes.fi)
Irma Virjo (irma.virjo@uta.fi)

Version: 5 Date: 26 January 2005

Author's response to reviews:

Dear Editorial staff at the BMC - Medical Informatics and Decision Making,

We gratefully thank you for accepting our manuscript - and also for the all distinguished comments during the revision process. We have checked typing errors, corrected the tables and formatted them using the "Excell Table object". Besides our final version of "USE OF EMAIL FOR PATIENT COMMUNICATION IN STUDENT HEALTH CARE" which we kindly ask to be published in the BioMed Central Medical Informatics and Decision Making.

Sincerely,

Johanna Castren, PhD Student at the University of Tampere